Author's response to reviews

Title: Improvements In Survival Of The Uncemented Nottingham Total Shoulder Prosthesis: A Prospective Comparative Study

Authors:

Nahum Rosenberg (nahumrosenberg@hotmail.com)
Lars Neumann (larsneumann@rcsed.ac.uk)
Amit Modi (dramitmodi@yahoo.com)
Istvan J Mersich (istvan_mersich@hotmail.com)
Angus W Wallace (angus.wallace@rcsed.ac.uk)

Version: 2 Date: 23 June 2007

Author's response to reviews:

Reply to reviewer 2:

Thank you for your comments. Accordingly we revised the following:

* We added the details on the postoperative rehabilitation program in the end of the Methods section.
* We added the details on the patients with rotator cuff tears in the Methods section and in the Table 2.
* There were no periprosthetic fractures that were treated conservatively. We have no exact data on the minor transient complications such as superficial infection or nerve injuries so those were not added to the text.

We hope this is sufficient for publication of our manuscript.

Sincerely
Nahum Rosenberg MD